%File content.tex, 9 September, 1992
%\magnification=\magstep1
\input ij.sty
\input mssymb
\def\ {\thinspace}
\lineskiplimit=100pt 
\lineskip=7pt %This controls the spacing between indexitems of
%both kinds
\tenpoint
\null\vskip 3cm

\def\index\indexvolume#1\indexyear#2\endindextitle{
\rm\parindent=0pt%\overfullrule=0pt
\centerline{ISRAEL JOURNAL OF MATHEMATICS}
\vskip0.4cm
\centerline{TABLE OF CONTENTS, VOLUME #1\unskip, #2\unskip}
\vskip1.2cm
\frenchspacing}
\def\endindex{}

\def\indexitem\indexauthor#1\indextitle#2\indexpage#3\endindexitem
{\halign {\normalbaselines%
\vbox{\hsize=11cm\hangindent=2em\hangafter=1 ##}%
&\hbox to 1.3cm{\hfill ##}\cr%
{\smc #1\unskip}\quad{\sl #2} & #3 \cr}\par}
\def\indexauthor{}
\def\endindexitem{}
\def\indexpage{}
\def\indextitle{}

\def\indexsee\indexauthor#1\partner#2\endindexitem
{{\smc #1\unskip\enspace}{\sl See }{\smc #2\unskip}\par}

\index
\indexvolume 113
\indexyear 1999
\endindextitle

\indexitem 
\indexauthor Alesker, S.
\indextitle On the measure of intersection of cylinders
\indexpage 231
\endindexitem

\indexitem
\indexauthor Barros, M., J.~L.~Cabrerizo and M.~Fern\'andez
\indextitle Reduction of\break variables for Willmore--Chen submanifolds
in seven spheres
\indexpage 29
\endindexitem

\indexitem
\indexauthor Brendle, J.
\indextitle Between P-points and nowhere dense ultrafilters
\indexpage 205
\endindexitem

\indexsee
\indexauthor Cabrerizo, J.~L.
\partner M.~Barros
\endindexitem

\indexitem
\indexauthor Cholak, P.~A. and A.~Nies
\indextitle Atomless $r$-maximal sets
\indexpage 305
\endindexitem

\indexitem
\indexauthor D\v{z}amonja, M. and S.~Shelah
\indextitle ${\clubsuit}$ does not imply the existence of a Suslin
tree
\indexpage 163
\endindexitem

\indexsee
\indexauthor Fern\'andez, M.
\partner M.~Barros
\endindexitem

\indexsee
\indexauthor Garc\'\i a-S\'anchez, P.~A.
\partner J.~C.~Rosales
\endindexitem

\indexitem
\indexauthor Gross, B.~H.
\indextitle Algebraic modular forms
\indexpage 61
\endindexitem

\indexitem
\indexauthor Loke, H.~Y.
\indextitle Dual pairs correspondences of $E_{8,4}$ and $E_{7,4}$
\indexpage 125
\endindexitem

\indexitem
\indexauthor Mao, Z. and S.~Rallis
\indextitle On a cubic lifting
\indexpage 95
\endindexitem

\indexitem
\indexauthor Matou\v{s} ek, J. and P.~Valtr
\indextitle On visibility and covering by convex sets
\indexpage 341
\endindexitem

\indexitem
\indexauthor Neumann, K.
\indextitle A criterion for the primeness of ideals generated by
polynomials with separated variables
\indexpage 1
\endindexitem

\indexsee
\indexauthor Nies, A.
\partner P.~A.~Cholak
\endindexitem

\indexitem 
\indexauthor Park, K.~K.
\indextitle On directional entropy functions
\indexpage 243
\endindexitem

\indexitem 
\indexauthor Petrogradsky, V.~M.
\indextitle Exponential generating functions and\break complexity of Lie
varieties
\indexpage 323
\endindexitem

\indexsee
\indexauthor Rallis, S.
\partner Z.~Mao
\endindexitem

\indexitem
\indexauthor Randrianantoanina, B.
\indextitle On isometric stability of complemented\break subspaces of
${\bf L}_p$
\indexpage 45
\endindexitem

\indexitem
\indexauthor Regev, A.
\indextitle Asymptotics of degrees of some $S_n$-sub regular
represen-\break tations
\indexpage 15
\endindexitem

\indexitem
\indexauthor Rosales, J.~C. and P.~A.~Garc\'\i a-S\'anchez
\indextitle Presentations for subsemigroups of finitely generated
commutative semigroups
\indexpage 269
\endindexitem

\indexsee
\indexauthor Shelah, S.
\partner M.~D\v{z}amonja
\endindexitem

\indexitem
\indexauthor Siburg, K.~F.
\indextitle Aubry--Mather theory and the inverse spectral problem
for planar convex domains
\indexpage 285
\endindexitem

\indexsee
\indexauthor Valtr, P.
\partner J.~Matou\v{s}ek
\endindexitem

\endindex
%End of file, content.tex
